Author’s response to reviews

Title: Cardiovascular risk factors and memory decline in middle-aged and older adults: The English Longitudinal Study of Ageing

Authors:

Beatriz Olaya (beatriz.olaya@pssjd.org)

Maria Victoria Moneta (mvictoria.moneta@pssjd.org)

Martin Bobak (m.bobak@ucl.ac.uk)

Josep Maria Haro (jmharo@pssjd.org)

Panayotes Demakakos (p.demakakos@ucl.ac.uk)

Version: 4 Date: 04 Nov 2019

Author’s response to reviews:

Editor Comments:

Our apologies for any confusion. We notice that some authors are now missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

After specifying the individual contributions of each other, please then include a statement to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

RESPONSE= Thank you. We have specified the individual contribution of each author and added the statement that all authors have read and approved the final manuscript.